Author's response to reviews

Title: Clinical presentation and predictors of outcome in patients with severe acute exacerbation of chronic obstructive pulmonary disease requiring admission to intensive care unit

Authors:

Alladi Mohan (alladimohan@yahoo.com)
Raya Premanand (prema2@sancharnet.in)
Lebaka N Reddy (lebaka2003@yahoo.co.in)
Mangu H Rao (drmhrao@yahoo.com)
Surendra K Sharma (sksharma@aiims.ac.in)
Ranjit Kamity (ranjith.kamity@gmail.com)
Srinivas Bollineni (shsri_b@yahoo.co.in)

Version: 2 Date: 4 July 2006

Author's response to reviews:

There is one typographical error in the first (title page) of the article (3564300821078721).

In the lines
"From:
The Division of Pulmonary and Critical Care Medicine, Department of Medicine, Department of Anesthesiology, Sri........"

there should be an asterix (*) before "Department of Anesthesiology...." and these lines should read

"From:
The Division of Pulmonary and Critical Care Medicine, Department of Medicine, *Department of Anesthesiology, Sri........"

I am resubmitting the manuscript with this correction.